Author’s response to reviews

Title: Analysis of Total Knee Arthroplasty revision causes in a tertiary care hospital in Germany

Authors:
Anne Postler (anne.postler@uniklinikum-dresden.de)
Cornelia Lützner (cornelia.luetzner@uniklinikum-dresden.de)
Franziska Beyer (franziska.beyer@uniklinikum-dresden.de)
Eric Tille (eric.tille@uniklinikum-dresden.de)
Jörg Lützner (joerg.luetzner@uniklinikum-dresden.de)

Version: 3 Date: 08 Feb 2018

Author’s response to reviews:

Thank you for reviewing our paper again. In detail we have done the following changes.

1) A list of all the abbreviations used in the manuscript has been added.

2) The Ethics approval and consent section of the Declarations was completed.

3) The data availability statement was included.

4) The Authors contributions section of the Declarations were revised and include now the details of all authors individually.

5) Figure files contain just the image now without any titles nor legends.